Author's response to reviews

Title: Regional differences in mortality in Greece (1984-2004): the case of Thrace

Authors:

Panagiotis PP Papastergiou (panpapast@med.uth.gr)
George GR Rachiotis (gsrachmed@yahoo.com)
Konstantina KP Polyzou (panpapast@med.uth.gr)
Christos CZ Zilidis (zilidis@gmail.com)
Christos CC Hadjichristodoulou (xhatzi@med.uth.gr)

Version: 2 Date: 19 November 2007

Author's response to reviews:

Dear Editor,

We resubmit our manuscript entitled `Regional differences in mortality in Greece (1984-2004): the case of Thrace` since a typographical error occurred in the previous cover letter. Descriptive analysis of mortality data can generate the hypothesis on possible risk factors associated with increased mortality rates in the region of Thrace. It is solely submitted to this Journal and all authors have read and approved the manuscript.

Yours sincerely,

Christos Chadjichristodoulou
Ass. Professor of Epidemiology
Department of Hygiene and Epidemiology
School of Medicine
University of Thessaly
e-mail: xhatzi@med.uth.gr